Author's response to reviews

Title: General practitioners' views and experiences of counselling for physical activity through the New Zealand Green Prescription program

Authors:

Asmita Patel (asmita.patel@aut.ac.nz)
Grant M Schofield III (grant.schofield@aut.ac.nz)
Gregory S Kolt (g.kolt@uws.edu.au)
Justin W L Keogh (jkeogh@bond.edu.au)

Version: 3 Date: 1 November 2011

Author's response to reviews:

Dear Editor

We have made the following changes to the format of the manuscript as requested by the BMC Series Editorial Production Team (both the major revisions and the minor revisions).

1. The list of authors in the manuscript is written exactly as they are in the submission system.

2. The highlighting/tracking has been removed from the manuscript.

3. The title has been written in sentence case.

4. The table placement text has been removed.

5. The "Authors' Information" section has been removed.

6. The manuscript has been checked for typographical errors and several small errors corrected.

Kind regards
Gregory S Kolt, PhD